Author’s response to reviews

Title: Implementation of infection control in health facilities in Arua district, Uganda: a cross-sectional study

Authors:

Peter Wasswa (pwasswa@afenet.net)
Christine K. Nalwadda (cnalwadda@musph.ac.ug)
Esther Buregyeya (eburegyeya@musph.ac.ug)
Sheba N. Gitta (sgitta@afenet.net)
Patrick Anguzu (yanguzu@yahoo.com)
Fred Nuwaha (fnuwaha@musph.ac.ug)

Version: 5
Date: 4 June 2015

Author’s response to reviews: see over
Response to Editors’ comments

Manuscript Title: Implementation of infection control in health facilities in Arua district, Uganda: a cross-sectional study

Date: 25 May 2015

1. Has the questionnaire used been previously published or is publicly available? If not, please indicate if an English language version is available and provide this if possible.

   Response: The questionnaire has neither been published before nor is it publically available. An English version has been submitted to the Editor.

2. Please present the box on page 10 as a figure rather than within the main manuscript.

   Response: The box has been presented as a figure and submitted separately.

3. Please clarify in the acknowledgements whether there were any sources of funding for the study.

   Response: A statement has been added to the acknowledgments stating that “The authors declare that there were no external sources of funding for this study and all costs incurred were self-sponsored.”

4. Please ensure the references are formatted in journal style. For example, Citations in the reference list should include all named authors, up to the first six before adding 'et al.'

   Response: The references have been formatted as per the journal style.